Dear Colleague,

Thanks for the time spent in reviewing our paper. Also thanks for your appreciation of our work. All your suggestions have been taken care adequately as follows:

1. A short table (Table 2) summarising the research findings has been added. The changes have been highlighted in red. Other changes have been made according to the other Reviewer’s advice.

Dear Colleague,

Thanks for the time spent in reviewing our paper. Also thanks for your appreciation of our work. All your suggestions have been taken care adequately as follows:

1. We think that there are 2 or 3 more case reports. Unfortunately, as stated by you, their analysis is hampered by the fact that they are not written in English. It would be an obstacle for the readers interested in the field as well. As such, we have excluded any papers with language other than English (selection criteria).
2. The echocardiogram views (4- and 2-chamber views) have been described in Figure 1 legend now.
3. A short table (Table 2) summarising the research findings has been added.

The changes have been highlighted in red. Other changes have been made according to the other Reviewer’s advice.